Author’s response to reviews

Title: Diet satisfaction and associated factors among adult surgical orthopaedic patients admitted at a teaching hospital in Lusaka province, Zambia; A hospital-based cross-sectional study

Authors:
Nixon Miyoba (nixonmiyoba@gmail.com)
Irene Ogada (irene.a.ogada@gmail.com)

Version: 5 Date: 30 Jan 2019

Author’s response to reviews:

Kitwe Teaching Hospital,
Nutrition and Dietetics Unit,
P.O Box 20969,
Kitwe.
30th January 2019

BMC Nutrition
Dear Editor,

NUTN-D-17-00206R4

Diet satisfaction and associated factors among adult surgical orthopaedic patients admitted at a teaching hospital in Lusaka province, Zambia; A hospital-based cross-sectional study

Nixon Miyoba, Master of Science - Food, Nutrition and Dietetics; Irene Ogada, Doctor of Philosophy - Nutrition and Dietetics; Jonathan Mulenga, Medical Doctor

BMC Nutrition
Dear Editor,

Editor Comments:
Comment No.1: Please include a title page to your manuscript that is properly formatted according to our submission guidelines. Response No.1: The title page has been properly formatted according to the journal guidelines.

Comment No. 2: Please add a heading "Declarations" after your List of abbreviations before the References section. Response No. 2: The declarations section has been placed in the appropriate position as advised.

Comment: No. 3: Please provide more justification for the contributions of authors. We ask that you consider your list of authors as it currently stands with reference to our guidelines regarding qualification for authorship. Response No. 3: The third person does not qualify for authorship based on the guidelines and his name has since been dropped from the authorship list.

Yours faithfully,

Nixon Miyoba